Author's response to reviews

Title: Induction of beta Defensin 2 by NTHi requires TLR2 Mediated MyD88 and IRAK-TRAF6-p38MAPK Signaling Pathway in Human Middle Ear Epithelial Cells

Authors:

Haa-Yung Lee (hlee@hei.org)
Tamotsu Takeshita (ttake@hama-med.ac.jp)
Jun Shimada (jasmd513@aol.com)
Arsen Akopyan (aakopyan@hei.org)
Jeong-Im Woo (jwoo@hei.org)
Huiqi Pan (hpan@hei.org)
Sung K Moon (smoon@hei.org)
Ali Andalibi (aandalibi@hei.org)
Rae-Kil Park (rkarpark@wonkwang.ac.kr)
Sung-Ho Kang (kkdin@kku.ac.kr)
Shin-Seok Kang (skang@hei.org)
Robert Gellibolian (rgellibolian@hei.org)
David J Lim (dlim@hei.org)

Version: 4 Date: 13 June 2008

Author's response to reviews: see over
June 12, 2008

BMC Infectious Diseases
Editorial Office

Re: 4056955151765301

Dear Sir/Ma’am:

I would like to submit the reformatted manuscript for publication in the Journal of BMC Infectious Diseases. We have reformatted the manuscript to reflect all queries and suggestions made by the reviewers.

The details are as follows:

1. Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals → Changed to small letter.
2. Country - Please remember to include the country in the affiliation details → All countries are included.
3. Email addresses - Please include each email address on a new line → Changed to each email address on a new line.
4. Materials and Methods - Please rename this section ‘Methods’ → Changed to ‘Methods’.
5. Conclusions - Please insert a Conclusions section after the Discussion. We recommend including a Conclusions heading above the paragraph which beings ‘In conclusion’ → Changed according to the suggestion.
6. Author’s contribution - Please change this to "Authors' contributions" and state that all authors read and approved the final version of the manuscript → Changed according to the suggestion.

I do hope that you will find the revised manuscript acceptable for publication. Should you have any questions, please do not hesitate to contact me using the information provided below.

Sincerely,

David, J. Lim, M.D.
Gonda Department of Cell and Molecular Biology
House Ear Institute
2100 W, 3rd Street
Los Angeles, CA 90057
Tel: 1-213-353-7021
Fax: 1-213-989-6778
e-mail: dlim@hei.org